

# **VU Research Portal**

#### Effect of spatial adaptation measures on flood risk in the coastal area of Flanders

Koks, E.E.; de Moel, H.; Bouwer, L.M.

2012

document version Publisher's PDF, also known as Version of record

Link to publication in VU Research Portal

*citation for published version (APA)* Koks, E. E., de Moel, H., & Bouwer, L. M. (2012). *Effect of spatial adaptation measures on flood risk in the coastal area of Flanders*. (Report; No. R12-12). Institute for Environmental Studies.

General rights Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
- You may not further distribute the material or use it for any profit-making activity or commercial gain You may freely distribute the URL identifying the publication in the public portal ?

#### Take down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

E-mail address: vuresearchportal.ub@vu.nl

# Effect of spatial adaptation measures on flood risk in the coastal area of Flanders

E.E. Koks<sup>1</sup>, H. de Moel<sup>1</sup>, L.M. Bouwer<sup>1</sup>

1 Institute for Environmental Studies, VU University Amsterdam

Valorisation Report 10 August 2012 This report is released by: Laurens Bouwer





This report was commissioned by: CCASPAR project (IWT agreement 080043)

IVM Institute for Environmental Studies VU University Amsterdam De Boelelaan 1087 1081 HV AMSTERDAM T +31-20-598 9555 F +31-20-598 9553

E info@ivm.vu.nl

#### **Copyright © 2012, Institute for Environmental Studies**

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system or transmitted in any form or by any means, electronic, mechanical, photo-copying, recording or otherwise without the prior written permission of the copyright holder

# Contents

Sumn	nary	5
1	Introduction	7
2	Study Area: Coastal Flanders	9
2.1 2.2 2.3	Socio-economic characteristics Land-use characteristics Geographic characteristics	9 10 11
3	Future developments	12
3.1 3.2	Socio-economic scenarios Climate change	12 13
4	Methodology	16
4.1 4.2 4.3	Land-use model Inundation model Damage model	17 19 21
5	Results	25
5.1 5.2 5.3 5.4	Baseline situation (2005) Future scenarios Spatial adaptation measures Flood damage with ports included	25 26 28 34
6	Discussion	38
6.1 6.2	Feasibility spatial adaptation measures Methodological issues	40 41
7	Conclusion	43
Refer	ences	45

#### Summary

Flood risk in coastal zones can increase due to climate change induced sea-level rise and socioeconomic changes. Over the last decades, population growth, increases in wealth and urban expansion have been found to be the main cause for increasing losses in coastal zones. These changes may, however, be offset by appropriate adaptation measures. The main goal of this study is therefore to assess the effectiveness of flood risk adaptation measures for the Flemish coastal zone, considering future climate and land-use change. In order to achieve this, we set up a modelling framework to assess the flood risk of the Flemish coast including climatic and socio-economic projections, and used this model to test the effectiveness of two spatial adaptation measures; compartmentalization and land-use zoning. In this modelling framework a land-use model, an inundation model and a damage model were combined to calculate the flood risk, in terms of expected annual damage, for those different situations.

The results show that without adaptation measures, future flood risk will increase between 40-50% by the year 2040. Of that increase, between 30-40% is the result of socio-economic changes, compared to an approximate 6% increase due to climate change. Only when considering a scenario of extreme sea-level rise does climate change increase the flood risk more than socioeconomic change.

Compartmentalization resulted in an average risk reduction of approximately 50% for both the baseline situation and for future scenarios. With the exception of the most extreme climate change scenario, compartmentalization successfully offsets the combined adverse effects of socio-economic growth and climate change on flood risk for this case study. Land-use zoning resulted in lower risk reductions compared to compartmentalization as it can only reduce the increase in risk. Land-use zoning resulting in reductions of future flood risk averaging between 6-10%. When only looking at the increase in risk, land-use zoning could offset about 15-20% of the increase in flood risk due to land-use change. For both compartmentalization and zoning, large differences have been found in their effectiveness between different locations.

Flood risk can also be reduced by strengthening the primary defences. In this study flooding starting from return periods of 100 years has been considered. If primary defences would be improved to withstand any water level with return periods up to 1000 years, flood risk will be reduced by almost 80%. Clearly, a variety of measures should be investigated for each area, in order to judge which measures are most effective in which location. Combining feasibility and costs could then result in an optimal combination of measures to manage flood risk for the Flemish coast.

Lastly, it should be noted that port areas constitute a large part of the total flood risk. In our calculations the flood risk in port areas actually surpassed that of the embanked parts. Despite this high potential damage, we did not test adaptation measures in this area because the investigated adaptation measures are not well suited for port areas and the known difficulty with estimating flood damage to industrial areas, which are much more heterogeneous with specialised buildings and equipment as opposed to residential areas. Therefore, more research has to be done to correctly simulate flood risk in these areas and consequently be able to test appropriate adaptation measures for ports.

## 1 Introduction

Coastal zones, such as the coastal zone of Belgium at the North Sea, are one of the most vulnerable areas to flooding (e.g. Nicholls et al., 2010). Flood risk in coastal zones will increase due to climate change and sea level rise (Gaslikova et al., 2011; Ponsar et al., 2007; Philippart et al., 2007). Furthermore, continuous socioeconomic growth in these coastal regions increases exposure to coastal flooding (McGranahan et al., 2007). Over the last decades, population growth, increases in wealth and urban expansion have been found to be the main cause for increasing losses in zones at risk from flooding (Bouwer et al., 2007; Nicholls et al., 2008).

These projected trends have led to research to assess what measures may be developed to reduce flood risk. However, accurate flood risk assessments for future developments and quantification of the effect of possible adaptation measures are often a complicated task and surrounded by uncertainties. Various studies have attempted to combine projections of climate change and socioeconomic trends to estimate future flood risk. These studies vary from assessment on a national or European scale (e.g. Hall et al., 2005; Klijn et al., 2012; Feyen et al., 2009), to studies focused on a regional scale (e.g. Maaskant et al., 2009; Bouwer et al., 2010; Mokrech et al., 2008; te Linde et al., 2011). These studies, however, mainly focused on the assessment of future flood risk and few looked into possible adaptation measures to lower this risk. There have been some studies that quantified the effect of adaptation measures on flood risk. These include for instance Bouwer et al. (2010) for a dike ring area in the Netherlands and De Kok and Grossmann (2010) in the Elbe region.

For the coast of Flanders, there have been various studies into baseline and future flood risk (Kellens, 2011; Van der Biest et al., 2009a; Verwaest et al., 2009) and the effect of possible adaptation measures (Van der Biest et al., 2009b). All studies took different approaches to calculate this risk, but none took different future socioeconomic scenarios into account and the specific adaptation measures that have been tested in this study.

The main goal of this study, therefore, is to assess effectiveness of flood risk adaptation measures for the Flemish coastal zone, considering future climate and land-use change. In order to achieve this goal, the research objectives are to set up a flood risk model for the Belgium coast and use this model to test the effectiveness of flood management measures in terms of risk reduction. This will be done in a few steps: (1) potential flood events will be simulated using an inundation model based on De Moel et al. (2012).(2) With these simulated flood events, potential flood damage will be calculated using a damage model based on methods developed by De Bruijn (2006) and Klijn et al. (2007). (3) Inundation depths and potential flood damage will be assessed for the baseline situation, as well as under future climate-change (sea-level rise) and land-use change scenarios. Finally, (4) two types of spatial adaptation measures, compartmentalization of the coastal area and land-use zoning, will be implemented in the model to compare them with the baseline situation to see how effective they are in mitigating flood risk in the Belgian coastal zone.

These activities, performed within WP1 of the CcASPAR project, are strongly linked with WP5 (the coastal case study), which addresses spatial adaptation measures for, amongst others, flood risk using 'research by design'; where compartmentalization of the coastal region is a key theme.

This report is organized as follows. Chapter 2 provides an overview of the case-study area. Chapter 3 provides a description of methods used with a description of the flood

#### 8 Introduction

risk model, the socio-economic- and climate scenarios and the potential measures Chapter 4 describes the results and Chapter 5 provides a discussion and recommendations Chapter 6, concludes.

# 2 Study Area: Coastal Flanders

The Belgian coast, the most western part of Flanders, is situated in the southern part of the North Sea between The Netherlands and France. The low-lying coastal area has a length of approximately 65km and an average width of 20km (Verwaest et al., 2007). The coastal region can be defined, according to Provoost and Hoffman (1996), as "an ecological coherent and functional area, consisting of marine environment, beaches, mud flats, salt marshes, dunes and polder areas". For this analysis, we base our coastal area on the sixteen municipalities that are defined in Van der Biest et al. (2008) as coastal municipalities<sup>1</sup> (Figure 2.1). In this chapter, we will briefly discuss the socioeconomic, land-use and geographic characteristics of the Flemish coastal zone.



Figure 2.1 Coastal zone of Flanders (in dark grey)

#### 2.1 Socio-economic characteristics

The coastal area of Flanders is a region with a relatively high population density compared to the rest of Flanders, respectively 678 inhabitants per km<sup>2</sup> versus 306 inhabitants per km<sup>2</sup> (Breyne et al., 2007). Due to the limited length of the coastline and the increasing population pressure, most of the coast has become urbanized. With around 120.000 inhabitants, Bruges (Figure 2.2) is the largest city in the region and with around 65.000 inhabitants is Oostende the largest city on the Flemish coastline (Kellens et al., 2011). In total, approximately 400.000 people live in this region (Mertens et al., 2010).

The main sectors of economic activity for the coastal region are tourism, agriculture and port industry. The coastal municipalities attract millions of tourists every year, resulting in a large amount of seasonal work in spring and summer. In 2007, the tourism in the coastal region has generated a turnover of approximately 2.6 billion Euros and 13.000 direct jobs (Volckaert and De Sutter, 2009). The downside, however, is a lot of housing vacancy in the low season during the winter (Breyne et al., 2007).

Due to the presence of highly fertile polder soil, the coastal plains of Flanders feature substantial amounts of farmland. The region of West-Flanders accounts for around 35% of the Flemish agricultural area, playing an important role in the Flemish agricultural

<sup>&</sup>lt;sup>1</sup> The following municipalities are included (from south to north): De Panne, Veurne, Kokszijde, Nieuwpoort, Middelkerke, Gistel, Oostende, Oudenburg, Bredene, Jabbeke, De Haan, Zuienkerke, Brugge, Blankenberge, Damme and Knokke-Heist.

sector. In the coastal area, about 40% of the farmland constitutes of pastures and grassland, whereas the other 60% is mainly used as arable land to cultivate grain crops (Belpaeme and Konings, 2004).

Moreover, main industry 'hubs' in the region are the ports of Zeebrugge and Oostende, both specialized in roll-on-roll-off (short: RoRo) port activities. In 2006, Zeebrugge was the largest RoRo-port of Europe handling more than 1.9 million new cars. In Oostende, 78% of the total throughput (6.2 million tonnes) was roll-on-roll-off. Zeebrugge also plays a key role in the European gas network and is a large distributer of containers (Breyne et al., 2007). In 2007, the ports together accounted for 7.7% of the total value added produced in Flanders. Other industrial activities are spread across the region and are generally concentrated in local small and medium enterprises (Van der Biest, 2008).

#### 2.2 Land-use characteristics

The land use of the coastal area of Flanders is diverse. One the hand, we see a relatively high building density in cities such as Oostende and Zeebrugge. On the other hand, we see a large amount of nature area and farming land. Directly on the coast, the high building density is for a large part the result of the large amount of holiday accommodation and second houses. The industry in the region is mostly located near the ports of Zeebrugge/Bruges and Oostende, with a few small exceptions in Nieuwpoort and Veurne.

The core nature is primarily located directly near the coast, in and around the dunes and beaches, whereas the recreational nature is mainly located directly near the cities and villages. More inland, the land use further away from the villages and cities is mostly farmland. However, the type of farmland varies throughout the region. More in the south-western part of the region, we find mainly regular cropland and pastures. In the northern part of the region, there is much more cropland with a nature function. The most common cultivated crops in this area are corn and grain (Van der Biest, 2008).



Figure 2.2. Land use of Coastal Flanders in 2005

When focussed more specifically on the ports of Zeebrugge and Oostende, we see that the port of Oostende is much more diverse. Whereas the main industrial land use in the port of Zeebrugge is Roll-on Roll-off Industry, is the main industrial land use in Oostende 'other industry'. This 'other industry' can be defined as a clustered land use of relatively small and diverse companies. Another interesting difference is the clustering of the port activities as a whole. In Zeebrugge, we see a relative high clustering of industrial activities, with a few (clustered) residential areas. In Oostende, on the other hand, we see much more residential areas in between the industrial areas.



Figure 2.2. Land use in the Ports of Zeebrugge and Oostende in 2005

#### 2.3 Geographic characteristics

As previously been said, the coast of Flanders has a length of approximately 65km and an average width of 20km. The elevation of most of the coastal zone is about two meters below the annual storm surge level of 5,5m TAW (Verwaest et al., 2005) and without the appropriate coastal protections, this area would flood every year (Mertens et al., 2010).

As we can see in Figure 2.4, the digital elevation map of the Belgian coastal zone, most of the coastal area is relatively flat. There are, however, a couple of exceptions. Near Bruges, we see the start of the higher sandy and sandy loam soils, which mark the end of the low-lying coastal zone. In the south-west, we clearly see the dunes between De Panne and Nieuwpoort. This thick strip of dunes is part of a natural coastal defence against floods. More to the north, we see again a similar strip of dunes between Oostende and Blankenberge and near Knokke. All the way up to the north, there is a tidal inlet at the border with the Netherlands, called the Zwin (Mertens et al., 2010). Moreover, other higher parts of the coastal zone that can be seen in Figure 2.4 can be defined as built up areas in and around the coastal cities.



Figure 2.4. Digital Elevation Map of Coastal Flanders

# 3 Future developments

When one wants to correctly measure the effectiveness of spatial adaptation measures, a valuable tool is the use of scenario analysis. Scenarios provide alternative views of the future, which can be used to test the influence of, for example, adaptation measures (Carter et al., 2001). In this report, we will be using different socio-economic and climate scenarios to measure the effect of spatial adaptation measures for the Belgian coastal zone in the future. As described by Berkhout et al. (2001), climate impact assessments require a clear picture of both climate and socio-economic changes. Combining the two makes it is possible to evaluate the exposure to harm from climate change and how future societies may adapt to these impacts. To do this, the socio-economic scenarios will be used as input for the land-use model to simulate several land-use projections for Flanders, whereas the climate scenarios will be used to determine different possible inundation levels in the coastal area.

#### 3.1 Socio-economic scenarios

In the second valorisation report (VR2) of the CcASPAR project (De Moel et al., 2012), we set up four socio-economic scenarios for the future of Flanders. These scenarios are used to develop four future land-use maps of Flanders. These scenarios are based on the quadrants of the two axes developed by SRES (IPCC, 2000) and used, amongst others, by WLO in the Netherlands (WLO, 2006). These two axes define the scale at which society makes decisions and policy (national vs. international) and the involvement of central government to steer developments (public vs. private). The narratives are in broad lines similar to those as in other studies, but are made Flanders specific. Below a brief explanation of each scenario:

#### Global Economy (A1)

The Global Economy scenario is characterized by a relatively weak governmental influence and strong globalisation. The Global Economy scenario has the largest growth in population and economy, resulting in the largest pressure on the finite amount of land in Flanders. Because of scaling up processes and increased global trade, agricultural land use is the least competitive and is the main component making space for other components. Environmental awareness is relatively low and nature is not seen as an important land use.

#### Transatlantic Market (A2)

In the Transatlantic Market scenario, there is a relatively little governmental interference, but unlike the A1 scenario, there is a more regional focus. Population growth is relatively low in this scenario compared to A1 due to less immigration, but because of strong reduction in the size of households, there is still a decent demand for new residential land use. The regional character of this scenario puts agriculture at a higher priority because of the wish to remain (largely) self-sufficient in food production. Combined with low environmental consciousness, this means that natural areas are under strong pressure to convert to other types. Areas of cultural heritage are protected up to a certain degree because of the regional focus of this scenario.

#### Strong Europe (B1)

In contrast to the A2 scenario, the Strong Europe scenario is characterized by strong governmental influence and globalisation. This globalisation results, via immigration, in a strong demand of urban land use, though the reduction household size is not as

big as in the A-scenarios. Urban development is, however, much more steered to combat urban sprawl and develop compact cities. Also designated natural areas are much better protected, though this is less the case for cultural heritage. Like in the A1 scenario, agricultural land use is the least competitive and most likely to decrease.

#### Regional Communities (B2)

The Regional Communities scenario denotes strong governmental influence and a regional focus. Like in the B1 scenario, environmental consciousness is large in this scenario. This result in the protection of natural areas (VEN), but also areas of cultural heritage are considered important. The amount of urban development is the least in this scenario, because of low population growth, low immigration and a limited reduction in household size. The focus of new nature development is on the combination with agriculture in order to limit foreign dependency for food security.

Figure 2.3 show the future land-use projections, based on the four socio-economic scenarios, for the coastal area of Flanders. In the A-scenarios, we see a typical pattern of urban sprawl occurring and a strong increase in the typical Flemish ribbon development, as expected in these scenarios. For coastal Flanders, this pattern is the most visible in the outskirts of Bruges, where new residential areas are build in former pastures and cropland. In contrast, the B-scenarios show much denser residential areas, which is expected with more strict policy rules. Moreover, we see in the A-scenarios a large increase in recreational nature areas. This increase is not only in the coastal areas at the expense of core nature areas, but also at the expense of pastures and cropland around the larger cities.



Figure 3.1. Land use of Coastal Flanders in 2040, according to the four different scenarios. Clockwise starting in the upper left corner: A1, B1, A2 and B2.

In all scenarios we see an increase in built up area, ranging from a large increase in the Global Economy scenario to a relative small increase in the Regional Communities scenario. Still, in all future scenarios we see a stronger pressure on the available land, varying from the need for residential area in the more trade-oriented scenarios, to the need to preserve nature area in the somewhat more self-sufficient scenarios. For an extensive explanation of the scenarios, we refer to the second valorisation report (De Moel et al., 2012).

#### 3.2 Climate change

The last decades, we've seen a change in the earth's climate and a rise in global mean sea level. From 1961 to 2003, there was an average rate of sea level rise between 1.3 and 2.3 mm  $yr^{-1}$  (Bindoff et al., 2007). For the Belgian coast in particular, we observed

#### 14 Future developments

the last century a mean sea-level rise of approximately 1.7 mm yr<sup>-1</sup> (Ozer et al., 2008). As it is expected that climate change and sea-level rise will continue (IPCC, 2007), the danger of flooding will increase for the Belgian coast. For the coast of Flanders, a mean sea level rise of at least 60cm can be expected by 2100 (Ponsar et al., 2007). Consequently, this can result in an increased flood probability due to higher sea levels, waves and storm surges.

At present, some parts of the coastal defence system of the Belgian coast can utmost withstand a storm surge with a probability of occurrence of once every 100 years (Verwaest et al., 2008). In the integrated Master Plan for Coastal Safety, however, it was stated that the government of the Flemish region defined a minimum safety level of the coastal defence at once in every 1000 years. Unfortunately, this safety standard is not implemented in any law or decree (Mertens et al., 2008). More than half of the Flemish coast is protected against flooding by reinforcement measures, such as dikes, jetties, weirs, locks and quay walls (Belpaeme and Konings, 2004). Nevertheless, some weak links have been identified along the Flemish coast line (Van der Biest et al., 2009). In every harbour and in several coastal cities, the quay walls and dyke levels are not able to withstand extreme storms, resulting in an overflow or breach during storm surges (Mertens et al., 2011). See Figure 3.2 for the possible breaches in the Belgian coast that will be used in this study.



Figure 3.2. Weak links for the Belgian coast. From left to right: Nieuwpoort, Ravenszijde, Bredene, Blankenberge and Knokke-Heist.

To estimate future changes in flood probabilities along the coast of Flanders, we used two climate scenarios, based on a moderate and worst case scenario defined in the CLIMAR project for the Belgian coast (Ponsar et al., 2007). These scenarios are calculated for both 2040 and 2100. Moreover, these two scenarios are similar to the ones used in Van der Biest et al. (2009).

	M – 2040	M – 2100	W – 2040	W – 2100
Air temperature	+ 1°C	+ 2°C	+ 2°C	+ 4°C
Sea water temperature	+ 1,2°C	+ 2,5°C	+ 1,7°C	+ 3,5°C
Change in air circulation	Yes	Yes	Yes	Yes
Change in wind speed	+ 2%	+ 4%	+ 4%	+ 8%
Change in winter precipitation	+ 7%	+ 14%	+ 14%	+ 28%
Change in summer precipitation	- 10%	- 20%	- 20%	- 40%
Average sea-level rise	30 cm	60 cm	50 cm	200 cm

Table 3.1	Climate	scenarios	for	coastal	Flanders	(after	2100)
-----------	---------	-----------	-----	---------	----------	--------	-------

As can be seen in Table 3.1, Scenario 'M' is a scenario with moderate changes in climate, whereas Scenario 'W' is a worst case scenario with extreme changes in temperature, precipitation and sea-level rise. These climate scenarios will be used to define the amount of water inflow per breach and the related inundation levels in the area.

# 4 Methodology

To analyze the effect of spatial adaptation measures for the Belgian coastal zone, a modelling framework is set up to assess their effects in both the current and future situations, taking possible future climate and socioeconomic changes into account. To do this, flood risks will be calculated for the current and possible future situations, with and without spatial adaptation measures. In general terms, flood risk can be defined as a combination of the probability of a flood event and its consequences (Kron, 2002; Samuels & Gouldby, 2005). More specifically; the flood risk is a function of the flood hazard's probability, the exposure and the vulnerability of the exposed socioeconomic system (Klijn et al., 2004, Samuels et al., 2006). Consequently, the flood risk will be expressed in terms of the expected annual damage (EAD) of flooding in a certain region, one of the common risk indicators used (Meyer et al., 2009; Ward et al., 2011). The expected annual damage can be defined as the sum of all possible flood risks or the integral below the probability-loss curve (see formula 1). To calculate this expected annual damage, we will be using a combination of a land-use model, inundation model and a damage model. For a consistent comparison between all the possible outcomes, a comparison will be made in terms of relative changes in flood risk. The comparison using relative changes is chosen as research shows that relative changes are more robust (Bubeck et al., 2011).

$$EAD = \sum_{i=1}^{n} \left( \frac{D_i - D_{i+1}}{2} \right) + (P_{i+1} - P_i) * D_{i+1}$$
(1)

Equation (1) with:

EAD = Expected Annual Damage n = number of flood events i = flood events D = Damage for flood event P = Probability for flood event

Figure 4.1 presents an overview of the modelling framework which is used in this study. To begin with, land-use maps and inundation maps will be created, which need respectively a land-use model and an inundation model. Following, a map with flood damages will be created with the use of the damage model, which combines the land-use maps and inundation maps with information of the vulnerability of the different land-use classes to flooding. Finally, with the calculated flood damages and the regarding flood probabilities, the Expected Annual Damage is calculated as defined in formula (1). The spatial adaptation measures will both be implemented on the input side of the model. Land-use zoning is expected to influence the land-use patterns (and will thus be implemented in the land-use model), whereas compartmentalization is expected to influence the inundation patterns of the area (and will thus be implemented in the inundation model).



#### 4.1 Land-use model

The land-use model that will be used for the coastal area is based on the Land Use Scanner for the whole of Flanders, which is a GIS-based land-use model that simulates future land use. The Land Use Scanner is developed to integrate and allocate future land-use demand from different sector-specific models or experts (Dekkers and Koomen, 2007; Hilferink and Rietveld, 1999). External regional projections of land-use change, which are usually referred to as demand or claims, are used as input for the model. These are land-use type specific and can be derived from, for example, sectorspecific models. The projected land-use changes are considered as an additional claim for the different land-use types as compared with the present area in use for each landuse type. The total of the additional claim and the present area for each land-use function is allocated to individual grid-cells based on the suitability of the cell. This local suitability may incorporate a large number of spatial datasets such as the current land use, physical properties, operative policies and market forces (generally expressed in distance relations to nearby land-use functions) (see Figure 4.2 for an overview of the model). For the future land-use projections, we use the four socioeconomic scenarios of Flanders, as described in Section 3.1. For a complete explanation how this model works and how the land-use maps are created, the reader is referred to VR2 (De Moel et al., 2012). However, in contrast to De Moel et al. (2012), only future projections are made for the province of West-Flanders. Consequently, claims used in De Moel et al. (2012) were recalculated from the whole of Flanders to West-Flanders specific. This made it possible to better steer the projections according to the future socioeconomic scenarios as described in Section 3.1. For this analysis, calculations are made for the years 2040 and 2100. For 2100, the same land-use maps are used as for 2040. The main reason for that choice is the high uncertainty of how land use will develop in the next 90 years (Klijn et al., 2012).

#### 18 Methodology





#### 4.1.1 Land-use zoning

Additionally, the model will be extended by implementing a spatial adaptation measure: land-use zoning. Land-use zoning can be defined as the concept of local planning and restrictive building of, for example, residential and industrial area. With more local planning and more restrictive building, there are possibilities to reduce the flood risk in a region. For example, instead of building in the areas with the highest inundation depths, new building sites are restricted to areas with relatively low or no inundation. Areas with a high flood probability and high inundation depths can, for example, better be used for nature of recreational land-use then for residential or industrial land-use.





The concept of zoning will be implemented in the land-use model by restricting certain areas for future residential, (non-)commercial and industrial development. This is done

by giving a negative local suitability to the areas defined in Figure 4.3. These restricted areas have either high flow velocities (i.e. areas very close to weak links in the coastal defence) or high inundation depths. Due to the restriction of building in these areas, we expect to reduce the consequences of floods and consequently the flood risk itself for coastal Flanders.

#### 4.2 Inundation model

The inundation model that is used to determine the amount of inundation in the coastal area is based on the inundation model developed by De Moel et al. (2012). As can be seen in Figure 4.4, first the hydraulic load, which is the water level during a storm surge event, and the size of the breach have to be determined. This is needed to estimate the volume of water flowing into the area after a breach. The amount of inflowing water depends on both the size of the breach and the difference in water level outside and inside the embankment. This difference in water level determines both breach growth and discharge through the breach. The material of the embankment itself also plays in important role in how fast the breach grows. The total amount of water which enters through the breach is calculated by integrating the discharge through the breach is calculated by integrating the discharge through the breach is calculated by integrating the discharge through the breach over time.

The next step, the inundation model itself, calculates the maximum inundation depth of an area after flooding, given the total volume calculated with the use of the hydraulic load and breach growth. The inundation model is mainly based on an inundation model, in this case the ´Digitaal Hoogte Model´, which has a resolution of 10m by 10m and gives elevation in centimetres above TAW. The first step in the model is to split the area into numerous individual micro-basins. In total, the coastal area was divided into thousands of micro-basins, varying in size between 100 m<sup>2</sup> for areas with a lot of relief and 15 km<sup>2</sup> or more for large flat polder areas. After this, the total volume is divided over the micro-basins, starting with the micro-basin next to the breach location and subsequently filling micro-basins until the total volume is met. For a more detailed description of the inundation model, the reader is referred to De Moel et al. (2012). Important to note, is that this model is only being used for 'inner dike' areas. For the port areas (e.g. Zeebrugge), this flood simulation was not possible, because they have an open connection to the sea. Therefore, these areas are entirely inundated by the maximum water level of the storm surge.



Figure 4.4 Basic layout of the Inundation model, based on the framework developed in De Moel et al. (2012)

In this study, several inundation scenarios have been used to simulate possible floods in coastal Flanders, based on Willems (2007). In Figure 4.5, the blue line represent the used return periods for the baseline situation in this study, varying from flood

probabilities of 1/100 up to 1/100000. To extrapolate these water heights to future scenarios, we added the amount of sea-level rise (in centimetres) to the probability curve of the baseline situation, resulting in higher water levels for the different return periods.



Figure 4.5 Overview of the different maximum water heights per return period for the baseline scenario.

#### 4.2.1 Compartmentalization

Compartmentalization can be defined as the splitting of a 'system' in multiple parts. In the case of flooding, compartmentalization can be used to diminish the surface area that will be flooded due to a single flood event. Consequently, by reducing the surface area that will be flooded, the consequences of a flood will be smaller and the flood risk will reduce (Klijn et al., 2010). The most convenient way of implementing this adaptation measure is to use, for example, old dikes, roads, railways, and other line elements that are already present. This is also in line with various other studies (Alkema and Middelkoop, 2005; Oost and Hoekstra, 2009; Theunissen et al., 2006) Often, these structures are already higher in the landscape and can thus be relatively easy transformed into secondary water defence structures. In this study, two types of compartmentalization will be tested. In the first compartmentalization measure (which will be named 'Compartments'), we raised each line element up to the maximum current height of the element to close the line element. By doing this, we created for the whole coastal region enclosed compartments. For the second compartmentalization measure (which will be named 'Compartments Plus'), we raised the created line structures of the first compartmentalization measure by another one meter.

To implement the concept of compartmentalization into the model, two new digital elevation maps have been created with closed/elevated line elements. In these new digital elevation maps, we made use of line structures that are already elevated in the map. See Figure 4.6 for an overview of the used line elements.



Figure 4.6 Digital Elevation Map of Coastal Flanders with compartmentalization

#### 4.3 Damage model

The final part of the calculations is to estimate the flood damage in the area. This can be calculated with the input maps that are created in the previous two models. To determine the amount of flood damage, we created a damage model which is based on the Damage Scanner. This model has originally been developed by De Bruijn (2006) and calculates the damage per land-use class, with the use of a land-use and inundation map. These maps are then combined using depth-damage curves and maximum damages per land use class. For each land-use class, there is a different maximum amount of possible damage and a different damage function. These depthdamage functions indicate the vulnerability by relating the inundation depth to a damage factor, which constitutes the fraction of the maximum damage that will be sustained at that inundation level (Koks et al., 2012). For a more detailed description of the Damage Scanner the reader is referred to De Bruijn (2006), Bouwer et al. (2009) and De Moel and Aerts (2011).

The maximum damages and damage functions were created using various sources. As can be seen on the land-use map in Figure 2.2, maximum damages and damage functions need to be defined for the thirteen land-use classes that are used in this study. First, we will derive the maximum damages based on the maximum damages defined in various other studies (e.g. de Bruijn, 2006; Klijn et al., 2007; Koks et al., 2012; Vanneuville et al., 2006). After that, we can define the depth-damage curves that are needed to correctly assess the flood damage that corresponds to a specific inundation depth.

#### 4.3.1 Maximum damages

Maximum damages have been defined for each land-use class considered by the landuse model (see Table 4.1). The values are based on existing models and studies from the Netherlands (Kok et al., 2005; Tebodin, 1998, 2000), the United Kingdom (Penning-Rowsell et al., 2010), Belgium (Vanneuville et al., 2006) and the United States (FEMA, 2007). For the calculation of maximum damages for the residential land uses, industrial sites and (non-)commercial land use, maximum damages from the Multi-Coloured Handbook (Penning-Rowsell et al., 2010), HAZUS damage model (FEMA, 2009) and Dutta et al. (2003) have been applied into this study. For each of these landuse classes, three types of damages are used: the maximum damage of the structure of the building, the maximum damage of the content of the building and the maximum damage of the area between each building (open space). To get a maximum damage for a land-use type, a weighted sum is taken from these three maximum

#### 22 Methodology

damages. The share of building/content/open space is used as weight and taken from a high resolution land-use map (Top10vector).

For (non)-commercial services, however, this approach was not as straight forward as for residential and industrial land use. A similar approach is conducted, but with the addition of taking the share of each type of land use within this land-use class into account (i.e. commercial, cultural, sports and governmental land use). This resulted in a weighted average for the maximum damage of these different land uses. For the different port industries, the maximum damages are based on studies from Tebodin (1998, 2000). Finally, the maximum damages for the different types of natural land use (e.g. core nature and forest) are set to zero, since no economic value can be attached to these areas. This is consistent with studies of Briene et al. (2002), Trouw et al. (2005) and Vanneuville et al. (2006).

	Land-use class	Max damage per m <sup>2</sup> (in Euros)
1	Residential – high density	560
2	Residential – medium density	300
3	Residential – loose buildings & ribbons	300
4	Roll-on Roll-off industry	490
5	Container terminals	540
6	Break bulk cargo	690
7	Bulk terminals	340
8	Chemical industry	390
9	Gas industry	390
10	Transport Industry	490
11	Food industry	490
12	Other port industry	490
13	Industrial sites	740
14	(non-)Commercial services	750
15	Horticulture	65
16	Cattle farming	0.1
17	Agriculture	0.5
18	Nature – Agriculture	0.5
19	Nature – Forestry	0
20	Nature – Recreation	0.03
21	Core nature	0
22	Infrastructure	40
23	Water	0

#### 4.3.2 Depth-damage curves

In this study, depth-damage curves are being used to calculate the flood damage per land-use for a specific flood. Each damage function used is defined by its land-use class. Residential land use, industrial sites and (non-)commercial land use have the same depth-damage curve, derived from the Multi-Coloured Handbook (Penning-Rowsell et al., 2010), HAZUS model (FEMA, 2009) and Dutta et al. (2003).

As can be seen in Figure 4.7, the 'Built-up Area' curve grows fairly quickly up to 1.5 meter. After that, it slows down up to three meters of inundation, where after it shortly grows much quicker again. The reason for this is that we expect that around three meters of inundation, we arrive at the first floor of the building, where new content

and parts of the structure will be destroyed (De Moel et al., submitted). The depthdamage curve for horticulture is based on the assumption that glass green houses will get more easily destroyed than the concrete buildings in the normal built-up area, but slower than agricultural land use. For this depth-damage curve, we assume that up to half a meter the damage increases fairly quickly, but after that it will increase somewhat slower until it will reach the maximum damage around 3.5 meters of inundation.  $\notin/m2$ 



Figure 4.7 General depth-damage curves for the damage model

Infrastructure is expected to increase fairly linear, following Kok et al. (2005). Cattle farming, agriculture and nature-agriculture are all clustered in the 'agriculture' curve, based on Koks et al. (2012) and Hoes (2007). Since crops and grass grow often very near the ground, it is expected that most of the damage will already occur in low inundation depths. For recreation, the curve is based on depth-damage curves created by Vanneuville et al. (2005). It is assumed that after a low inundation depth, no extra damage will occur. The only costs that will occur are clean-up cost. For the other nature land uses and water, no depth-damage curves are created, as it is expected that no damage occurs to these land uses. For the different industrial land-use classes in ports, the depth-damage curves are based on Tebodin (1998, 2000). See Figure 4.8 for an overview of the used depth-damage curves for these classes.

### 24 Methodology



Figure 4.8 Industrial depth-damage curves for the damage model

### 5 Results

In this chapter, we will describe the results of the analyses. We will start with briefly describing the damage calculations for the reference situation (the land-use map for the year 2005 combined with floods in the baseline situation). These reference calculations are then used to compare the results for the future land-use maps, the future climate scenarios and the spatial adaptation measures. Finally, results will be described where ports are taken into account.

#### 5.1 Baseline situation (2005)

For the reference situation, flood damages are calculated by using the land-use map of the year 2005 and flood events with return periods as shown in Figure 4.5 (i.e. return periods from 1/100 up to 1/100,000). Figure 5.1 shows an example of the results. As can be seen in the figure, a storm with a probability of 1/100,000, most of the flood damage occurs around Oostende. Interestingly, the highest inundation depths can be found near Blankenberge, with depths up to 125 centimeters for most of the inundated area. This is in contrast to the other breaches, where often only a small part of the area has large inundation depths. However, as most of the damage near Blankenberge occurs outside the built-up area, the damage is rather low.





Table 5.1 shows the total results for the baseline situation. The total expected annual damage for the coast of Flanders is approximately 25 million euro per year. As can be seen from the fourth column (EAD for residential area), almost all of the damage occurs in residential area. The agricultural area clearly has the lowest share in the total damage, with less than 0.02 million Euro per year. The breach south west of Oostende has the highest share of damage, taking up almost 60% of the total damage. This is related to a large amount of built-up area and a relatively large inundated area. Interesting is to compare the total EAD with the average area inundated area per breach. The average area inundated is the arithmetic mean over the inundated area of the eight flood events used in this study. Even though the southern breach near Oostende clearly has the highest share in expected annual damage, it has only a 1% higher share in the average inundated area compared to the second largest inundated area (the northern breach near Oostende). This is in contrast to a difference of three times in expected annual damage between the two breaches. Notable is also the large

share of industrial damage in Nieuwpoort, which makes up a quarter of the total damage there, much more than at the other breach locations.

Table 5.1 Expected Annual Damage for the baseline situation (in brackets the share of a	each
breach compared to the total number at the top of each breach)	

	EAD (10 <sup>6</sup> €/yr)		Av a inur (in	erage irea idated ikm <sup>2</sup> )	EAD for residential area (10 <sup>6</sup> €/yr)		EAD for industrial area (10 <sup>6</sup> €/yr)		EAD for infrastructure (10 <sup>6</sup> €/yr)		EAD for agricultural area (10 <sup>6</sup> €/yr)	
Total	25.4		1	. <b>7.2</b>	22.6	(89%)	2.1	(8%)	0.5	(2%)	0.02	(0.1%)
Nieuwpoort	2.1	(8%)	1.6	(9%)	1.58	(75%)	0.52	(25%)	0.03	(1%)	0.002	(0.10%)
Oostende SW	14.7	(58%)	5	(29%)	13.01	(89%)	1.3	(9%)	0.31	(2%)	0.002	(0.01%)
Oostende NE	4.9	(19%)	4.8	(28%)	4.65	(95%)	0.1	(2%)	0.1	(2%)	0.01	(0.22%)
Blankenberge	0.65	(3%)	3.1	(18%)	0.63	(97%)	0.01	(2%)	0.01	(2%)	0.002	(0.31%)
Knokke-Heist	2.93	(12%)	2.7	(16%)	2.74	(94%)	0.15	(5%)	0.04	(1%)	0.001	(0.03%)

As the circle diagram on the left hand side of Figure 5.2 shows, almost 50 percent of the inundated area is built-up area and more than 30 percent farmland. Even though the built-up areas take up 'only' 47 percent of the inundated area, the sum of these categories takes up almost 100 percent of the expected annual damage. This is in expense of the agricultural and nature areas, which is mainly the result of the much higher maximum damages for the built-up areas.



Figure 5.2 Amount of inundation per land use versus the EAD per land use in the baseline situation

#### 5.2 Future scenarios

Expected annual damages for the future scenarios are calculated using the estimated damages under future land-use change and changes in sea-level rise due to climate change. For each scenario, a significant increase in flood risk can be observed. As can be seen in Table 5.2, most of the increase in expected annual damage occurs due to socioeconomic changes rather than climate change. For the year 2040, climate change only results in a 5-7% increase in damage, while socioeconomic changes result in a 33-41% increase in expected damage. The M-scenario in 2100 shows similar results, which is not surprising given that the sea-level rise of 60cm is not too different from the W-scenario for 2040 (50cm). The W-scenario for 2100, however, has a much larger sea-level rise assumption (200cm) and shows completely the opposite. Here, climate change results in a 65% increase in damage, compared to a 33-41% increase in damage

due to socioeconomic changes. Regarding the differences in in the share of each breach location to the total EAD, the share of Oostende stays similar for all scenarios except for the worst-case scenario in 2100. In the worst-case scenario for 2100, the share of Oostende in the EAD decreases with almost 10%, while the other breach locations all increase with a couple percent. This indicates that a 2 meter sea-level rise results in a relatively low increase in flood damage for the Oostende region in comparison with the other breach locations.

	Socioeconomic scenario											
		Baseline (2005)		A1		A2		B1		B2		
	Baseline situation	25.3		35.8	(+41%)	34.6	(+37%)	35.5	(+40%)	33.6	(+33%)	
Climate												
Climate scenario	M-Scenario 2040	26.6	(+5%)	38.0	(+50%)	36.7	(+45%)	37.7	(+49%)	35.6	(+41%)	
	W-Scenario 2040	26.9	(+6%)	38.5	(+52%)	37.2	(+47%)	38.2	(+51%)	36.0	(+42%)	
	M-Scenario 2100	27.1	(+7%)	39.0	(+54%)	37.6	(+49%)	38.6	(+52%)	36.4	(+44%)	
	W-Scenario 2100	41.7	(+65%)	68.4	(+170%)	64.7	(+156%)	67.9	(+168%)	59.3	(+134%)	

Table 5.2 Expected Annual Damage (10 <sup>6</sup> €	€/yr) and percent change (between brackets), und	er
climate change, socioec	conomic change, and combination.	

When plotting the storm surge height versus the flood damage (Figure 5.3), it becomes clear that between 7.8 en 8 meters of inundation, a significant increase in damage occurs. This is in contrast to the storm surge heights between 6.5 and 7.8 meters and between 8 and 9.5 meters, where only a relatively slow increase in flood damage is observed. This implies that after a storm surge height of 7.8 meters, the inundated area increases considerably. Results show that this is indeed the case for the breaches near Oostende and Blankenberge.





The increase in damage due to socioeconomic changes is mainly the result of increases in residential areas. For all the scenarios, the increase in residential areas account for between 65-70% of the increase in flood risk. Similar to the baseline situation, this is again at the expense of agricultural and natural area. Figure 5.4 illustrates the contribution to the change in expected annual damages from climate

change only (red), socioeconomic change only (green) and the combination of both impacts (purple) for scenario A1. The base risk (blue) reflects the situation for he baseline (2005). As can clearly be seen from the figure, the influence of the socioeconomic change is for the 2040-scenarios and the M-scenario for 2100 significantly larger than changes in climate, while for the W-Scenario in 2100 the influence of climate change is significantly larger. Even though this graph only shows the changes for the A1-Scenario, the contributions of both socioeconomic and climate change are for the other socioeconomic scenarios similar.





#### 5.3 Spatial adaptation measures

#### 5.3.1 Compartmentalization

As described in section 4.2.2, two different compartmentalization measures will be tested in this study. In the first compartmentalization measure ('Compartments'), each line element is raised to the highest elevation of that line element. By doing this, we created for the whole coastal region enclosed compartments. For the second compartmentalization measure ('Compartments Plus'), the created line structures of the first compartmentalization measure are raised by one meter. Figure 5.5 shows the result of both measures for the southern breach near Oostende and the breach near Blankenberge for a flood event with a probability of 1/2000 without climate change. As can be seen, the results differ a lot on a local level. At Oostende, the exposure significantly decreases to only a very small area due to compartmentalization. However, as can be seen from the figure, the inundation depth decreases and stays within a certain compartment. At Blankenberge, on the other hand, the inundation extent actually increases due to compartmentalization as floodwater is kept out of a deep polder area and has to find its way elsewhere. The main differences for these two cases are the size of the compartments (much more and smaller compartments around Oostende) and the height of the compartments (the original line structures around Oostende are in general higher). See Figure 4.6 for the overview of all the line structures that are used in this study.

#### Results



Figure 5.5 Difference in flooding due to compartmentalization for a flood with a 1/2000 probability

For the baseline situation both compartmentalization measures show a significant but more or less similar decrease in flood risk (Table 5.4). Both types of compartmentalization result in a more than 50% reduction in expected annual damage for the baseline situation (Table 5.4). Nevertheless, there are a lot of differences on a local level. For example, the EAD for the breach southwest of Oostende decreased by almost 100%, but increased by almost 80% for the breach northwest of Oostende. Even though the average area inundated increases by only 8%, almost all of this increase occurs in residential area, resulting in a large increase in damage. Interesting is the decrease of the expected annual damage for the breach near Blankenberge, despite the increase in inundated area (Figure 5.5). Due to compartmentalization, the floodwaters are directed around the village instead of through the village of Blankenberge. Although more (low-lying) agricultural land got inundated, the total damage reduced. Overall, the difference of the one meter extra elevation on the line elements (i.e. compartmentalization vs. compartmentalization plus) is rather limited.

	Baseline	e (2005)		'Compar	tments	,	'Compartments Plus'			
	EAD (10 <sup>6</sup> €/yr)	Average area inundated (in km <sup>2</sup> )	EAD (10 <sup>6</sup> €/yr)		Average area inundated (in km <sup>2</sup> )		EAD (10 <sup>6</sup> €/yr)		Average area inundated (in km <sup>2</sup> )	
Total	25.3	17.2	12.2	(-52%)	10.7	(-38%)	12.5	(-51%)	10.3	(-40%)
Nieuwpoort	2.1	1.6	2	(-5%)	1.1	(-31%)	2	(-5%)	1.1	(-31%)
Oostende SW	14.7	5	0.1	(-99%)	0.2	(-96%)	0.4	(-97%)	0.2	(-96%)
Oostende NE	4.9	4.8	8.7	(+78%)	5.2	(+8%)	8.7	(+78%)	5.2	(+8%)
Blankenberge	0.7	3.1	0.4	(-43%)	3.5	(+13%)	0.5	(-29%)	3.3	(+6%)
Knokke-Heist	2.9	2.7	1	(-66%)	0.8	(-70%)	0.9	(-69%)	0.5	(-81%)

Table 5.4 Expected Annual Damage (10 <sup>6</sup> €/yr) and average inunda	ated area for the baseline
situation with compartmentalization per breach	location.

Another option is to compare the compartmentalization measures using lossprobability curves. These curves show all the potential flood damages and their probabilities, including the extreme events. By collating all the flood damages, together with their flood probabilities, a loss-probability curve can be constructed (Figure 5.6). As can be seen in the figure, the baseline curve shows the relation between the flood probability and the amount of damage for the year 2005. Due to compartmentalization, a downward shift of the curve can be observed, indicating that less flood damage occurs for each flood event. Interestingly, even though the expected annual damages for both compartmentalization measures are similar, the 'Compartments' measure has a much higher flood damage for the most extreme flood event (p = 1/100,000). However, as the 'Compartments Plus' measure has slightly larger damages for the higher probability flood events (on average less than 1% higher), the EAD for both measures are eventually similar.



Figure 5.6 Loss-probability curves for the baseline situation with and without compartmentalization measures.

As described in Chapter 4, the compartmentalization measures are implemented in the inundation model. Thus, the measures will affect the result from the different flood events and therefore they will also be clearly visible in the future climate scenarios. For the future scenarios, Table 5.5 and Table 5.6 show the results. As can be seen in the tables, indeed a much lower flood risk is observed for the different scenarios compared to the future scenarios where no compartmentalization is implemented. When looking only to the effect of climate change, three out of four scenarios show an actual reduction in flood risk of about 50%, compared to the 5-7% increase in the 'no adaptation'- situation (see Table 5.2). For the worst-case scenario in 2100, the reduction because of compartmentalization is much smaller, around 11%, indicating that this measure becomes less effective with really high sea-level rise. Compared to the baseline risk, the risk-reducing effect of compartmentalization would more than offset the adverse effect of sea-level rise on the flood risk for both 2040 scenarios and the M-2100 scenario (13.3-14.2 vs. 25.3 million Euros per year). The W-2100 scenario, on the other hand, would still see an increase in absolute flood risk (37.2 vs. 25.3 million Euros per year). The results for the 'Compartmentalization Plus' measure under current land-use (Table 5.6) are very similar to those of the 'Compartmentalization' measure. The most notable difference is that the risk reduction for the W-2100 scenario is much larger for the 'Compartmentalization Plus' scenario than for the 'Compartmentalization' scenario. The risk-reducing effect is there much more in line with the other climate scenarios (-36%). This implies that for high sea-level rise, extra elevation of the line elements is necessary to effectively counteract the increased flood risk.

When looking at the flood risks in the future scenarios where climate change and socioeconomic changes are combined, the results show that compartmentalization

would more than offset the increase in flood risk due to socio-economic change. For both 2040 and the M-2100 scenarios, both compartmentalization measures would roughly half the risk as opposed to without the measure. Under these climate change scenarios, compartmentalization would even decrease the flood risk below the baseline level despite the risk increase because of both climate change and socioeconomic change. Only under the W-2100 scenario, the combined effect of climatic and socio-economic change is not offset by the compartmentalization measures. Nevertheless, the compartmentalization measures still have a considerable risk reducing effect under the W-2100 scenario, around 25% for 'Compartmentalization' and around 50% for 'Compartmentalization Plus'. This again shows that to combat very high sea-level rise, line elements need to be extra elevated with respect to their current maximum elevation.

Table 5.5 Expected Annual Damage  $(10^6 \notin /yr)$  and percent change (between brackets) for the 'Compartments' measure under climatic and socio-economic change. The change percentages are with respect to the same socio-economic and climate-change scenario without adaptation (see Table 5.2).

	Socioeconomic scenario											
		Baseline (2005)	A1	A2	B1	B2						
	Baseline situation	12.2 (-52%)	16.5 (-54%)	15.8 (-54%)	16.3 (-54%)	15.7 (-53%)						
Climato												
connario	M-Scenario 2040	13.3 (-47%)	19.0 (-50%)	18.3 (-50%)	18.7 (-50%)	18.1 (-49%)						
scenario	W-Scenario 2040	13.7 (-46%)	19.5 (-49%)	18.8 (-49%)	19.2 (-50%)	18.6 (-48%)						
	M-Scenario 2100	14.2 (-44%)	20.1 (-48%)	19.3 (-49%)	19.8 (-49%)	19.1 (-47%)						
	W-Scenario 2100	37.2 (-11%)	51.2 (-25%)	48.8 (-25%)	51.3 (-25%)	46.1 (-22%)						

see Table 5.2)

Table 5.6 Expected Annual Damage (10<sup>6</sup> €/yr) and percent change (between brackets) for the 'Compartments Plus' measure under climatic and socio-economic change. The change percentages are with respect to the same socio-economic and climate-change scenario without adaptation (see Table 5.2).

	Socioeconomic scenario												
	Baseline (2		Baseline (2005)		A1 A2		2	B1			B2		
	Baseline situation	12.5	(-51%)	16.2	(-55%)		15.5	(-55%)	16	.0	(-55%)	15.4	(-54%)
Climata													
cimate	M-Scenario 2040	13.5	(-47%)	17.7	(-54%)		17.0	(-54%)	17	.4	(-54%)	16.8	(-53%)
Scenario	W-Scenario 2040	13.8	(-46%)	18.1	(-53%)		17.3	(-53%)	17	.8	(-53%)	17.1	(-52%)
	M-Scenario 2100	14.1	(-44%)	18.5	(-53%)		17.8	(-53%)	18	.2	(-53%)	17.5	(-52%)
	W-Scenario 2100	26.5	(-36%)	35.6	(-48%)		33.9	(-48%)	35	.6	(-47%)	32.4	(-45%)

From Figure 5.7 it becomes quite clear that the 'Compartments' measure results in an overall smaller reduction in flood risk than the 'Compartments Plus' measure. For all the future scenarios there is clearly a lower flood risk due to the 'Compartments' measure compared to the 'Compartments Plus' measure. For the worst-case scenario in 2100, the differences are the largest. Although the flood risk doesn't actually decrease compared to the baseline situation, it is clear that the 'Compartments Plus' measure decreases the flood risk the most. Next to that, also the influence of the different socioeconomic scenarios becomes clear. For example, the A1-scenario, which has the highest socioeconomic increases, also has the least reduction in flood risk due to compartmentalization. This is in contrast to the B2-scenario, with the lowest socioeconomic increases. This smaller decrease in flood risk for the A1-scenario can

be explained by the larger amount of built-up area in the areas at risk, compared to the B2-scenario. Therefore, the adaptation measures have less effect.



Figure 5.7 Percentage change in flood risk for each scenario in the areas where compartmentalization is implemented

#### 5.3.2 Zoning

As described in section 4.1.2, land-use zoning will be implemented in the land-use model by restricting certain areas for future residential, (non-)commercial and industrial development. These restricted areas are areas that have either high flow velocities (i.e. areas very close to weak links in the coastal defence) or high inundation depths. Due to the restriction of building in these areas, it is expected that the consequences of floods will lower and consequently also the flood risk itself for coastal Flanders. See Figure 5.8 for an example how land use has changed in coastal Flanders.



Figure 5.8 Example of land-use zoning near Oostende in the A1-scenario The left panel is a simulation without zoning, and the right panel with the zoning restriction.

Because land-use zoning only restricts future expansion of built-up area, the adaptation measure will not influence the baseline situation. As described in Chapter 4, land-use zoning is implemented in the land-use model. Thus, the measure will affect

the future socioeconomic scenarios and not the flood extents and depths. Table 5.8 shows the results of land-use zoning. Overall, land-use zoning results in a much smaller decrease than compartmentalization. For the future socioeconomic changes, there is a decrease of 4-6% in flood risk compared to the same socioeconomic scenario without adaptation. For the flood risks in the future scenarios where climate change and socioeconomic changes are combined, the future scenarios for 2040 and the Mscenario for 2100, flood risks reduce between 4-6% compared to the same future scenario without adaptation measures. However, when only looking at the increase in risk, land-use zoning could offset about 15-20% of the increase in flood risk due to land-use change. For the worst-case scenario in 2100, the reduction is somewhat higher, between 6-11%. Due to the larger flood extent in this scenario, more areas that are restricted for new urban developments are now flooded compared to the other climate scenarios. This results in a relatively larger effect of the land-use zoning measure. When looking at the different land-use types which are restricted for future building (e.g. residential, industry, commercial and non-commercial land-use), the highest decrease can be seen in residential land-use.

	Socioeconomic scenario											
		Baselin	e (2005)	A	1	А	.2	B1		B2		
	CS	2	25.3		(-4%)	32.8	(-5%)	34.0	(-4%)	31.7	(-6%)	
Climate												
Climate .	M-Scenario 2040	26.6	(0%)	36.4	(-4%)	34.7	(-6%)	35.9	(-5%)	33.5	(-6%)	
scenario	W-Scenario 2040	26.9	(0%)	36.9	(-4%)	35.1	(-6%)	36.4	(-5%)	33.9	(-6%)	
	M-Scenario 2100	27.1	(0%)	37.3	(-4%)	35.5	(-6%)	36.7	(-5%)	34.2	(-6%)	
	W-Scenario 2100	41.7	(0%)	62.2	(-9%)	57.9	(-10%)	60.5	(-11%)	55.6	(-6%)	

Table 5.7 Expected Annual Damage (10<sup>6</sup> €/yr) and percent change (between brackets), under climate change, socioeconomic change, and combination for land-use zoning.

As can be seen from both Table 5.8 and Figure 5.9, the B2-scenario has a similar decrease in flood risk for all climate scenarios, whereas the other scenarios have a much higher decrease in flood risk for the worst-case scenario in 2100 compared to the other three climate scenarios. The main reason is the difference in amount of built-up area with and without land-use zoning. In the A1-, A2- and B1-scenario without adaptation, there is overall more built-up area in the coastal zone than in the B2-scenario. Thus, when implementing restrictions on urban development, there will be a relative larger decrease in urban areas. Consequently, this results in a larger decrease in flood risk.





Table 5.8 shows the results for the A1-scenario and B2-scenario in the M-2040 climate scenario per breach location. As can be seen from the table, there are differences between the two scenarios and even larger differences between each breach location. For example, in the A1-scenario the largest decrease can be found near Blankenberge (18% decrease compared to 'no adaptation'-situation), which is the result of a 15% decrease in residential flood damage. Contrastingly, no risk reduction is found at Blankenberge for the B2 scenario, implying that zoning had no effect on urban development there in the B2 scenario. Also for Nieuwpoort, no reduction in risk is found in the B2 scenario due to zoning. This can be explained by the fact that in the B2-scenario without adaptation, there was already no new built-up area compared to the baseline situation. At the other locations zoning does have an effect, which is most pronounced at the breach location northeast of Oostende, where zoning results in a 16% decrease compared to the 'no-adaptation'-situation, which is the result of a 90% decrease in industrial flood damage.

Table 5.8 Expected Annual Damage (10<sup>6</sup> €/yr) and percent change compared to specific scenario without adaptation (between brackets) for the different breach locations in the M-2040 climate scenario.

		A1		B2				
	No adaptation	Zor	ning	No adaptation	Zo	ning		
Total	38.0	36.4	(-4%)	35.6	33.5	(-6%)		
Nieuwpoort	3.0	2.9	(-3%)	2.7	2.7	(0%)		
Oostende SW	19.9	19.4	(-3%)	18.6	18.2	(-2%)		
Oostende NE	8.4	7.7	(-8%)	8.1	6.8	(-16%)		
Blankenberge	1.1	0.9	(-18%)	0.9	0.9	(0%)		
Knokke-Heist	5.7	5.4	(-5%)	5.3	4.9	(-8%)		

#### 5.4 Flood damage with ports included

Finally, results will be discussed where the ports of Oostende and Zeebrugge are included into the damage calculation. As can be seen in Figure 5.10, for a storm with a probability of 1/100,000, most of the flood damage occurs now around the ports of Oostende and Zeebrugge. This can be explained by two reasons.

#### Results



Figure 5.10 Flood damage for coastal Flanders in the baseline situation with a 1/100000 flood and ports included

First, due to the open connection with the sea, it is expected that both ports will be entirely inundated. Second, maximum inundation depths vary greatly between the port regions and the other breach locations, varying from 1 meter at the breach southwest of Oostende up to more than 5 meters in the port of Zeebrugge.

Table 5.9 shows the expected annual damage for the baseline situation. As can be seen in the table, more than 60% of the flood damage occurs in the port areas (as without ports included the EAD was 25.3 million euros per year). The main reason for this, as said in the paragraph above, is that the ports have an open connection with the sea. Therefore, the port areas will be flooded with higher inundation depths compared to the other breach locations, where there is a better protection from dikes or dunes. For ports, almost 70% of that flood damage occurs in the industrial areas. There are, however, also some residential districts outside the embankments, mainly at Oostende, which make up 29% of the flood damage there.

	EAD (	10 <sup>6</sup> €/yr)	Average area inundated (in km <sup>2</sup> )		EAD for residential area (10 <sup>6</sup> €/yr)		EAD for industrial area (10 <sup>6</sup> €/yr)		EAD for infrastructur e (10 <sup>6</sup> €/yr)		EAD for agricultural area (10 <sup>6</sup> €/yr)	
Total	68.6		3	1.7	35.1	(51%)	32	(47%)	1.4	(2%)	0.05	(0.1%)
Nieuwpoort	2.1	(3%)	1.6	(5%)	1.58	(75%)	0.52	(25%)	0.03	(1%)	0.002	(0.10%)
Oostende SW	14.7	(22%)	5.0	(16%)	13.01	(89%)	1.3	(9%)	0.31	(2%)	0.002	(0.01%)
Oostende NE	4.9	(7%)	4.8	(15%)	4.65	(95%)	0.1	(2%)	0.1	(2%)	0.01	(0.22%)
Blankenberge	0.7	(1%)	3.1	(10%)	0.63	(97%)	0.01	(2%)	0.01	(2%)	0.002	(0.31%)
Knokke-Heist	2.9	(4%)	2.7	(9%)	2.74	(94%)	0.15	(5%)	0.04	(1%)	0.001	(0.03%)
Ports	43.3	(63%)	14.5	(46%)	12.5	(29%)	29.9	(69%)	0.88	(2%)	0.03	(2%)

Table 5.9 Expected Annual Damage for the baseline situation, ports included

Even though the built-up areas take up 'only' 45 per cent of the inundated area, the sum of these categories takes up almost 100 per cent of the expected annual damage (Figure 5.11). This is in expense of the agricultural and nature areas, which is mainly the result of the much higher maximum damages for the built-up areas. Interesting is to compare Figure 5.11 with Figure 5.2, the land-use shares in inundation and EAD without ports. The total share of built-up area in the total inundation is for both cases more or less the same. However, instead of 45% of residential area there is now a 23% share in residential area and a 20% in port industry. For the share in EAD, a similar

observation can be found (90% residential area vs. 51% residential area and 39% port industry). The other land use shares stay in both figures almost the same.



Figure 5.11 Amount of inundation per land use versus the EAD per land use in the baseline situation

For the port areas, three land-use categories take up most of the annual expected damage: 'RoRo'-industry, container terminals and other industry (Figure 5.12). This is mainly due to two reasons. First, they take up a large share in the port areas (see Figure 2.2) and consequently a large share in the total inundated area (83%). Second, a large share of the Roll-on Roll-off industry and container terminals is located in the reclamation ground outside the port of Zeebrugge, resulting in high inundation depths and high damages because it will get struck first by the storm surge wave.





The share of the port stays similar for all scenarios except for the worst-case scenario in 2100. In the worst-case scenario for 2100, the share of ports in the EAD decreases with 10%, while the other breach locations all increase with a couple percent. This can be explained by the larger flood extent in the embanked areas of the coastal area for the worst-case scenario in 2100. Because the unembanked area of the ports is already severely damaged with relatively low inundation depths, the flood damage increases fairly linear. The other breach locations, however, have low flood damages with relatively low inundation depths and therefore will see a significant increase in flood damage when suddenly the storm surge height greatly increases. Consequently, this results in a decrease in share for the ports and an increase in share for the other breach locations. The increase in damage due to socioeconomic changes (Table 5.10) is mainly the result of increases in residential areas. Similar to the baseline situation and to the calculations without ports, this is still in expense of agricultural and natural area.

# Table 5.10 Expected Annual Damage (10<sup>6</sup> €/yr) and percent change (between brackets), under climate change, socioeconomic change, and combination.

	Socioeconomic scenario											
		Baseline	e (2005)	A1		A2		B1		B2		
	Baseline situation	68	68.6		(+34%)	87.3	(+27%)	89.6	(+31%)	85.9	(+25%)	
Climata												
Climate	M-Scenario 2040	73.3	(+7%)	98.0	(+43%)	93.1	(+36%)	95.5	(+39%)	91.4	(+33%)	
scenario	W-Scenario 2040	75.2	(+10%)	100.4	(+46%)	95.4	(+39%)	97.9	(+43%)	93.6	(+37%)	
	M-Scenario 2100	76.2	(+11%)	101.7	(+48%)	96.7	(+41%)	99.1	(+45%)	94.7	(+38%)	
	W-Scenario 2100	97.6	(+42%)	135.1	(+97%)	127.3	(+86%)	131.2	(+91%)	123.7	(+80%)	

# 6 Discussion

In this study, future flood risks are calculated for coastal Flanders by taking future socioeconomic and climate changes into account, using different scenarios. In total, four different socioeconomic scenarios and four different climate scenarios are used. Results show that due to climate and socioeconomic changes, flood risk in coastal Flanders will increase in the future, varying between 40 and 50% increases by the year 2040, compared to the baseline situation in 2005. Of that increase, between 30-40% is the result of socioeconomic changes, compared to approximately only a 6% increase due to climate change. Clearly, socioeconomic changes are the biggest influence for future increases in flood risk. This larger influence of projected socioeconomic changes on flood risk has also been observed in several other studies (Hall et al., 2005; Klijn et al., 2012; Feyen et al., 2009; Mokrech et al., 2008).

It is, however, doubtful if that many different scenarios are needed. Results are relatively similar between each socioeconomic scenario and between most of the climate scenarios. For example, the increase due to climate change varies between 5-7% for three out of four scenarios. The only real influence of climate change is in the worst-case scenario for 2100 where an increase in risk of more than 60% is observed. The socioeconomic scenarios differ even less. The difference between the two 'extremes' (A1 vs. B2) is less than 10%. This is, for example, in contrast to Hall et al. (2005), who state that there is a big difference between the socioeconomic scenarios used in the UK. The main reason for this small difference between the socioeconomic scenarios for the coastal area of Flanders. This is in contrast with, for example, Bouwer et al. (2011) where a large difference in socioeconomic scenarios is observed.

A few studies have calculated flood risk and flood damages for coastal Flanders (Kellens, 2011; Van der Biest et al., 2009a and 2009b; Verwaest et al., 2009). All studies took different approaches to calculate this risk, but none took different future socioeconomic scenarios into account, or the specific adaptation measures that have been tested in this study. For example, Kellens (2011) only looked at the flood risk for the baseline situation. When comparing the results of Kellens (2011) with this study, somewhat similar results can be observed when it comes to the distribution of the flood risk around the coastline. In both studies, Oostende and the ports represent the largest share in the calculated flood risk. The total risk, however, is different. In this study, for the baseline situation a flood risk is observed of around 25 million Euros per year without the ports. A similar result has been found in Kellens (2011), but in that study ports are included in this risk. When ports are included in this study, the total flood risk for the baseline situation is almost 69 million Euros per year.

Van der Biest et al. (2009a) only looked at one specific flood event for the baseline situation and two future scenarios. The future scenarios used in that study are similar with the scenarios in this study for the year 2100. Flood damages in Van der Biest et al. are calculated by using a 'super storm' event with a probability of 1/17,000. For the baseline situation, this resulted in a total flood damage of around 400 million Euros, according to their study. In this study, different storm surge heights are used with different return periods. Therefore, the flood damage of the 1/100,000 flood event will be used for comparisons, which has a somewhat similar storm surge height. In this study, the total flood damage for this event is around three billion Euros, almost eight times higher than in the study of Van der Biest et al. (2009). This can be explained on two grounds. First, in our study, a larger area was flooded for the baseline situation

#### Discussion

(approximately three times larger). Second, there are large differences in maximum damages and depth-damage curves, which may explain the remaining difference. These differences in flood exposure and in damage calculations, also explain the differences in damage between the future situations. Van der Biest et al. (2009b) looked into the effect of adaptation measures to climate change impacts. Most of these adaptation measures were focused on improved coastline protection (reducing the flood hazard). But one, however, was similar to the land-use zoning measure used in this study. Nevertheless, the results are different. For example, according to Van der Biest et al. (2009b), no specific restrictions are needed near Oostende. This is in contrast to this study, where it was found that around Oostende the largest decrease in flood risk can be achieved when implementing restrictions for future developments in built-up area.

Next to the calculation of future flood risks, two spatial adaptations measures are tested: compartmentalization by improving existing linear landscape elements and land-use zoning. Results show that compartmentalization will result in the highest decrease in future flood risk, and indicate a decrease of more than 50% compared to the future scenarios without adaptation measures. Land-use zoning, on the other hand, shows the smallest decrease in future flood risk, between 4-10%. However, when only looking at the increase in risk, land-use zoning could offset about 15-20% of the increase in flood risk due to land-use change. This large difference is mainly related to the fact that zoning only influences the flood risk of new urban developments, and does not influence the risk of the current building stock, contrary to compartmentalization that affects the flood extent, and thus also the risk of existing built-up areas. This has a large influence, as flood waters are kept away from existing large residential areas (see e.g. Oostende in Fig. 5.5).

However, there can be large differences for compartmentalization on a local level, varying from larger inundated areas to higher inundated depths (Alkema and Middelkoop, 2005; Oost and Hoekstra, 2009; Klijn et al., 2009). Those results are also found in this study. For example, even though the flood extent near the south of Oostende significantly reduced and resulted in a large decrease in expected annual damage, the flood extent near Blankenberge and north of Oostende increased due to compartmentalization. For Blankenberge, this still resulted in a decrease in expected annual damage due to the water now flooding the agricultural area behind the village of Blankenberge instead of the village itself. In the north of Oostende, on the other hand, there was an actual increase in expected annual damage. Even though the average area inundated increases by only 8%, almost all of this increase occurs in residential area, resulting in a large increase in damage.

In the Master Plan for Coastal Safety, it is stated that the Flemish coastline should be able to withstand at least a storm with a probability of 1/1000 (Mertens et al., 2008). In this study, we assumed that the weak spots of the coastline are still only able to withstand storms with a maximum probability of 1/100. However, if Belgium would increase its safety levels for the entire coastline to 1/1000, the total flood risk will decrease by almost 80%. This means that annual expected damages in the baseline situation will reduce from 25.3 million Euros per year, to approximately 5 million Euros per year. Compared to the tested spatial adaptation measures, this is a much larger reduction in flood risk, as the best spatial measure (compartmentalization) resulted in an average reduction of expected annual damages of around 50%. As current safety levels are still relatively low, it might be more effective for some locations to invest first in reducing the probability of flooding by raising and strengthening the primary coastal defences, instead of reducing the exposure and consequences by compartmentalization and land-use zoning, respectively.

The analysis has focussed on damages that occur in the coastal areas other than the ports. When taking the ports of Oostende and Zeebrugge into account, the baseline flood risk estimate increases by more than 170% compared to the flood risk for coastal Flanders without taking the damage in these ports into account. Clearly, these areas are very much at risk. Not only due to the open connection with the sea, but also because of the high economic activity and value of assets in these areas. However, due to the nature of the investigated spatial adaptation measures, it was not possible to apply them in these port areas. Port activities are almost by definition bound to unembanked areas, which means that compartmentalization is not a feasible option as that would result in the embankment of the area, hindering the daily activities. Also land-use zoning is limited in its potential in port areas because of the limited size and options for relocation there. Therefore, more research is needed into appropriate measures to reduce flood risk in those regions. Next to the problems with implementing the tested spatial adaptation measures in these regions, there is also the known difficulty with estimating flood damage to industrial areas, which are much more heterogeneous with specialised buildings and equipment as opposed to residential areas. Therefore, one has to be cautious with interpreting the damage results in these areas.

#### 6.1 Feasibility spatial adaptation measures

When integrating the tested spatial adaptation measures into flood risk management for a specific region, the economic feasibility to implement these measures has to be decided upon. For the coastal region of Flanders, results show that compartmentalization significantly reduces the flood risk. However, the effect of strengthening the primary defences in order to be able to withstand a storm with a probability of at least 1/1000 shows an even greater reduction in flood risk. Klijn et al. (2007) described a close relation between the failure probability of the primary defences and the benefits of compartmentalization. When the primary flood defences are still low, compartmentalization can deliver considerable benefits, but these might not be the measure with the highest possible benefits. However, when the primary flood defences are protected against very low probability storms (e.g. the strong flood defences along the Dutch coast), the benefits of compartmentalization will be also very small, as not much extra benefits can be achieved. Consequently, whether or not compartmentalization is effective should be assessed on a local level. As, for example, Figure 5.5 showed, large differences were found between different locations. For Blankenberge, it might be more beneficial if the primary defences will be upgraded to higher standards of protection. For Oostende, on the other hand, compartmentalization can reduce the flood risk significantly, without the need of considerable upgrades of the coastal defence, with detrimental effects of the current economic uses of the coastal fringe. By looking which measures is most effective in which area, an optimal safety level of protection can be reached that can be more effective in reducing risk than using a single measure in the entire area.

Nevertheless, compartmentalization is fairly easy to implement. In this study, old embankments and existing line structures (e.g. roads, railways and old dikes) have been used to test these measures. Therefore, no new dikes have to be build, only existing structures have to be strengthened. In the 'Compartments Plus' measure, these line structures have also been raised by one meter. This, however, only resulted in larger risk reductions compared to the 'Compartments' measure for the worst-case scenario in 2100, which is based on a sea-level rise of two meter.

#### Discussion

For land-use zoning, the assessment of benefits is more difficult. Future land-use change is uncertain and potential benefits of zoning are directly related to the amount of developments that can be restricted by zoning. If there is substantial socioeconomic growth, restrictions may reduce the potential of increased flood risk considerably. If there is, on the other hand, little to no growth, the potential to reduce flood risk by zoning is low and restricting developments is basically unnecessary. Therefore, land-use zoning might be an interesting option in combination with other spatial adaptation measures, but not as sole measure.

#### 6.2 Methodological issues

This analysis is not without its limitations. For starters, there are several biases around the creation of the future land-use maps, such as outdated information for the base map of 2005, uncertainty in the allocation of future land use and the linear extrapolation for the future projections. See De Moel et al. (2012a) for a more extensive explanation of these problems. Second, research shows that there are high uncertainties in the derived flood damages in flood risk analysis such as these. This is mainly the result of large uncertainties in the determination of the maximum damages and the depth-damage curves (De Moel and Aerts, 2011; Merz et al., 2004; Merz et al., 2009), but also the determination of the volume of water that could enter the polder area (De Moel et al., 2012). Therefore, one has to be cautious when interpreting these results. To somewhat correct for these uncertainties, all results are compared in relative terms.

Also, studies have showed that when calculating the expected annual damage, one has to be careful with selecting how much flood events to use for calculating it. When calculation the expected annual with damage with only very few flood events in the tail of the curve, the flood risk can be highly underestimated (Ward et al., 2011).

A final limitation was the inundation of the ports of Zeebrugge and Oostende. For these areas that are not protected by flood defences, the inundation model was not able to simulate individual inundation events. Therefore, the maximum water level of the storm surge was applied on the entire unembanked area. This may have resulted in an overestimation of inundation depths, as the effect of the shape of the harbour and its entrance were not accounted for. Furthermore, it should be noted that the inundation model used is a simplification of 2-dimensional hydraulic models, which can better capture hydraulic dynamics during a flood event.

## 7 Conclusion

In this study, the effect of climatic change, socio-economic change and two spatial adaptation measures on flood risk for the coastal area of Flanders have been investigated. In order to achieve this, we set up a flood risk model for the Flemish coast and used this model to test the effectiveness of two spatial flood management measures as how they reduce risk. This was done for four climate change scenarios and four socio-economic scenarios. A land-use model, an inundation model and a damage model were combined to calculate flood risk for coastal Flanders, which was quantified in terms of expected annual damage. The two spatial adaptation measures, compartmentalization and land-use zoning, were implemented in the inundation model and the land-use model, respectively. In this way, compartmentalization was assumed to limit the flood extent, and land-use zoning limits the (additional) exposure to flooding.

Results show that without adaptation measures, future flood risks will increase between 40-50% for the year 2040. Of that increase, between 30-40% is the result of socioeconomic changes, compared to an approximately 6% increase due to climate change. Climate change is assumed to affect only sea-level rise (i.e. inundation depth), and not the frequency of storm surges due to changes in extreme wind speeds during storms. Only when considering extreme sea-level rise does climate change increase the flood risk more than socioeconomic change. This is illustrated by the worst-case scenario for 2100 used in this study, which assumes a sea-level rise of two meter. When adaptation measures are being implemented, flood risks are substantially reduced. Compartmentalization resulted in an average reduction of approximately 50% of annual expected damages, for both the baseline situation as well as for future scenarios. Land-use zoning resulted in much lower reductions, averaging between 6-10%. Except for the most extreme climate change scenario, compartmentalization successfully offsets the combined adverse effects of socio-economic growth and climate change on flood risk. For both compartmentalization and zoning, large differences are found in their effectiveness for different locations. For example, compartmentalization significantly decreased the flood risk in most areas, while it actually increased in another area. Similarly, the effect of zoning also depends very much on the location, as well as on the socio-economic projection used. Large growth leads to larger avoided damages for the land-use zoning measure.

Flood risk can also be reduced by strengthening the primary defences. In this study flood events with return periods of 100 years and lower have been considered. If primary defences would be improved to withstand anything up to a return period of 1000 years, flood risk will be reduced by almost 80%. Clearly, a variety of measures should be investigated in various areas in order to judge which measures are most effective in which location. Combining feasibility and costs could then result in an optimal mix of measures to manage flood risk for the Flemish coast.

Lastly, it should be noted that port areas constitute a large part of the total flood risk. In our calculations the flood risk in port areas actually surpassed that of the embanked part. Despite this high potential damage, we did not test adaptation measures in this area. The main reason was the limitations of the inundation model to simulate floods correctly in these areas and the known difficulty with estimating flood damage to industrial areas, which are much more heterogeneous with specialised buildings and equipment compared to residential areas. Therefore, more research has to be done to

#### 44 Conclusion

correctly simulate flood risk in these areas and consequently be able to test adaptation measures.

#### References

- Alkema, D. and Middelkoop, H., 2005. The influence of floodplain compartmentalization on flood risk within the Rhine-Meuse delta. Natural Hazards 36.1-2: 125-45.
- Belpaeme K. & Konings Ph. (eds.), 2004. The Coastal Atlas Flanders-Belgium. Publication of the Co-ordination Center for Coatsal Zone Management, 100 pages.
- Berkhout, F., Hertin, J., Jordan, A., Socio-economic futures in climate change impact assessment : using scenarios as 'learning machines' Global environmental change, 12 (2, 2002) : 83-95.
- Bindoff, N.L., J. Willebrand, V. Artale, A. Cazenave, J. Gregory, S. Gulev, K. Hanawa, C. Le Quéré, S. Levitu, Y. Nojiri, C.K. Shum, L.D. Talley and A. Unnikrishnan, 2007: Observations: Oceanic Climate Change and Sea Level. In: Climate Change 2007: The Physical Science Basis. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change [Solomon, S., D. Qin, M. Manning, Z. Chen, M. Marquis, K.B. Averyt, M. Tignor and H.L. Miller (eds)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.
- Bouwer, L.M., Bubeck, P., Wagtendonk, A.J., and Aerts, J.C.J.H. (2009). Inundation scenarios for flood damage evaluation in polder area. Natural Hazards and Earth System Sciences 9: 1995-2007, doi:10.5194/nhess-9-1995-2009.
- Breyne, P.; Belpaeme, K.; Lescrauwaet, A.-K.; Mees, J.; Seys, J. (2007). "West-Vlaanderen door de zee gedreven": Rede door Paul Breyne, Gouverneur van West-Vlaanderen uitgesproken in de provincieraad van 2 oktober 2007. Provincie West-Vlaanderen: Brugge. 77 pp.
- Briene, M., Koppert, S., Koopman, A. and Verkennis, A. (2002). Financiele onderbouwing kengetallen hoogwaterschade. VROM, Rijkswaterstaat, Dienst Weg -en Waterbouwkunde (Dutch).
- Bubeck, P., de Moel, H., Bouwer, L.M., Aerts, J.C.J.H., 2011. How reliable are projections of future flood damage? Natural Hazards and Earth System Sciences, 11: 1-14.
- Burby, R. J., Beatley, T., Berke, P. R., Deyle, R. E., French, S. P., Godschalk, D. R., Kaiser, E. J., et al. (2007). Unleashing the Power of Planning to Create Disaster- Resistant. Journal of the American Planning Assocation, (65:3), 247-258.
- Carter, T.R., La Rovere, E.L., 2001. Developing and applying scenarios. In: McCarthy, J., et al. (Eds.), Climate Change 2001: Impacts, Adaptation and Vulnerability. Cambridge University Press, Cambridge, pp. 145-190.
- De Bruijn, K. M., 2006. Bepalen van schade ten gevolge van overstromingen voor verschillende scenario's en bij verschillende beleidsopties, WL | Delft Hydraulics, Delft, the Netherlands.
- De Moel, H., and Aerts, J. C. J. H., 2011. Effect of uncertainty in land use, damage models and inundation depth on flood damage estimates, Nat. Hazards, 58(1), 407-425, doi:10.1007/s11069-010-9675-6
- De Moel, H., Van Vliet, M. and Aerts, J.C.J.H., submitted. Evaluating the effect of flood damage-reducing measures: a case study of the unembanked area of Rotterdam, the Netherlands. Submitted to Regional Environmental Change
- De Klerck, P., 2011. Vergrijzing en ouderenzorg aan de kust: moet er nog (nieuw) zand zijn?. SVR-Webartikel (2011/8). Studiedienst van de Vlaamse Regering: Brussel. 13 p.
- Dekkers J.E.C., Koomen E. (2007) Land-use simulation for water management: application of the Land Use Scanner model in two large-scale scenario-studies (Chapter 20). In:

Koomen E, Stillwell J, Bakema A, Scholten HJ (eds) Modelling land-use change; progress and applications. Springer, Dordrecht, pp 355-373

- Dutta, D, S. Herath, and K. Musiake, A Mathematical Model for Flood Loss Estimation, Journal of Hydrology, Vol.277, 2003, pp.24-49.
- FEMA (2009) HAZUS-MH MR4 Flood Model Technical Manual, Federal Emergency Management Agency, Mitigation Division, Washington, D.C.
- Hilferink M, Rietveld P (1999) Land Use Scanner: an integrated GIS based model for long term projections of land use in urban and rural areas. J Geogr Sys 1(2):155-177
- Holway, J. M., & Burby, R. J., 1993. Reducing Flood Losses Local Planning and Land Use Controls. Journal of the American Planning Assocation, (59:2), 205-216.
- IPCC, 2000. Special report on emissions scenarios. Nakicenovic, N., R. Swart (eds.), Intergovernmental Panel on Climate Change, Cambridge University Press, Cambridge.
- IPCC, 2007. Climate change 2007; summary for policymakers. In: Climate change 2007: the physical basis, contributions of working group 1 to the fourth assessment report of the intergovernmental panel on climate change. Solomon, S., D. Qin, M. Manning, Z. Chen, M. Marquis, K.B. Averyt, M. Tignor and H.L. Miller (Ed.), p. 996, Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.
- Jacobs, C., Koomen, E., Bouwman, A. and Van der Burg, A. (2011). Lessons learned from land-use simulation in regional planning applications. Chapter 8 in: Koomen, E. and Borsboom-van Beurden, J. Land-use modeling in planning practice. Springer,
- Heidelberg.Kellens, W.; Neutens, T.; Deckers, P.; Reyns, J.; De Maeyer, P. (2011). Coastal flood risks and seasonal tourism: analyzing the effects of tourism dynamics on casualty calculations Natural Hazards Online First.
- Hoes, O.A.C. (2007). Aanpak wateroverlast in polders op basis van risicobeheer. Technische Universiteit Delft (Dutch).
- Maaskant, B., Jonkman, S.N., & Bouwer, L.M. (2009). Future risk of flooding: an analysis of changes in potential loss of life in South Holland (The Netherlands). Environmental Science and Policy, 12(2), 157-169.
- Oost, J. and Hoekstra, A.Y. (2009), Flood damage reduction by compartmentalization of a dike ring: comparing the effectiveness of three strategies. Journal of Flood Risk Management, 2: 315-321. doi: 10.1111/j.1753-318X.2009.01050.x
- Ozer, J., D. Van den Eynde and S. Ponsar, 2008. Trend analysis of the relative mean sea level at Oostende (Southern North Sea – Belgian coast). Technical Report CLIMAR/X/JO/200807/ EN/TR3, Report prepared in the framework of the CLIMAR project for the Belgian Science Policy, Contract SD/NS/01A, Management Unit of the North Sea Mathematical Models, Brussels, 13 pp.
- Penning-Rowsell, E. C., Viavattene, C., Pardoe, J., Chatterton, J., Parker, D., and Morris,
  J.:The benefits of flood and coastal risk management: a handbook of assessment
  techniques 2010, Flood Hazard Research Centre, Middlesex University Press, 2010.
- Philippart C.J.M., Anadon R., Danovaro R., Dippner J.W., Drinkwater K.F., Hawkins S.J., O'Sullivan G., Oguz T., Reid P.C., 2007. Impacts of climate change on the European Marine and coastal environment. European Sciences Foundation, Position Paper 9, pp 84
- Ponsar, S., J. Ozer and D. Van den Eynde, 2007. Impacts of climate change on the physical and chemical parameters of the North Sea (literature study). Report prepared in the framework of the CLIMAR project for the Belgian Science Policy, Contract SD/NS/01A, Management Unit of the North Sea Mathematical Models, Brussels, 70 pp.

- Provoost S. & M. Hoffmann (Ed.) (1996) Ecosysteemvisie voor de Vlaamse Kust: 1. Ecosysteembeschrijving. Ministerie van de Vlaamse Gemeenschap, Administratie Milieu-, Natuur-,Land- en Waterbeheer. Afdeling Natuur, Gent, België.
- Kellens, W. (2011). Analysis, perception and communication of coastal flood risks: Examining objective and subjective risk assessment = Analyse, perceptie en communicatie van overstromingsrisico's in kustgebieden: onderzoek naar objectieve en subjectieve risicobeoordeling. PhD Thesis. Universiteit Gent: Gent. ISBN 978-94-90695-82-8. 224 pp
- Kok, M., Huizinga, H.J., Vrouwenvelder, A.C.W.M. and Barendregt, A. (2005).
  Standaardmethode2004 Schade en Slachtoffers als gevolg van overstromingen. DWW-2005-005. 2005. RWS Dienst Weg- en Waterbouwkunde (Dutch).
- Koks. E.E., Moel, H. de., Koomen, E. (2012) Comparing extreme rainfall and large-scale flooding induced inundation risk- Evidence from a Dutch case-study. Chapter 1 in: Dr. Kumarasamy, M. (ed.) Studies on water management issues. Intech, Rijeka, (pp. 3-26)
- Klijn, F., M. van Buren, and S.A.M. van Rooij. 2004. Flood-risk management strategies for an uncertain future: Living with Rhine River floods in The Netherlands? AMBIO 33: 141-147.
- Klijn, F., Baan, P., de Bruijn, K. and Kwadijk, J. (2007). MNP. Overstromingsrisico's in Nederland in een veranderend klimaat. Verwachtingen, schattingen en berekeningen voor het project Nederland Later. WL - Delft Hydraulics (Dutch).
- Klijn, F., Asselman, N. and Van der Most, H. (2010), Compartmentalisation: flood consequence reduction by splitting up large polder areas. Journal of Flood Risk Management, 3: (pp. 3-17)
- Klijn, F., de Bruijn, K. M., Knoop, J., & Kwadijk, J. (2012). Assessment of the Netherlands' flood risk management policy under global change. Ambio, 41(2), 180-92. doi:10.1007/s13280-011-0193-x
- Kron, W. and Thumerer, T.: Water-related disasters: Loss trends and possible counter measures from a (re-)insurer's point of view Germany, 2002.
- Mertens, T., Trouw, K., Bluekens, K., De Nocker, L., Couderé, K., Sauwer, C., De Smedt, P., Lewis, C., Verwaest, T., (2008), SAFECoast: integrated master plan for flanders future coastal safety, Coastal Division of the Flemish Community, Belgium
- Mertens, T.; Verwaest, T.; Delgado, R.; Trouw, K.; De Nocker, L. (2011). Coastal management and disaster planning on the basis of flood risk calculations, in: (2011). The 32nd International Conference on Coastal Engineering (ICCE 2010), June 30 July 5, 2010, Shanghai, China: book of papers. (pp. 1-8)
- Merz B, Thieken AH (2009) Flood risk curves and uncertainty bounds. Nat Hazards 51(3):437-458.
- Merz B, Kreibich H, Thieken A, Schmidtke R (2004) Estimation uncertainty of direct monetary flood damage to buildings. Nat Hazards Earth Syst Sci 4(1):153-163.
- MIRA (2006) Milieurapport Vlaanderen, Achtergronddocument 2006, Kust & zee, Goffin A., Lescrauwaet A.-K, Calewaert J.-B., Mees J., Seys J., Delbare D., Demaré W, Hostens K., Moulaert I., Parmentier K., Redant F., Mergaert K., Vanhooreweder B., Maes F., De Meyer P., Belpaeme K., Maelfait H., Degraer S., De Maersschalck V., Derous S., Gheskiere T., Vanaverbeke J., Van Hoey G., Kuijken E., Stienen E., Haelters J., Kerckhof F., Overloop S., Peeters B., Vlaamse Milieumaatschappij, www.milieurapport.be
- Nicholls, R.J. & Cazenave, A. Sea-level rise and its impact on coastal zones. Science 328, 1517-1520 (2010).Nijkamp, P., Leeuwen, E.S. van & Rietveld, P. (2009). Climate change: From global concern to regional challenge. Romanian Journal of Regional Science, 3(2), 18-38.

- Tebodin (1998). Schade bij inundatie van buitendijkse industrie. Tebodin consultants & engineers. Document nr 334393. (In Dutch)
- Tebodin (2000). Schadecurves industrie ten gevolge van overstroming. Tebodin consultants & engineers. Document nr 334666. (In Dutch)
- Theunissen R., Kok M. & Vrijling H. Compartimentering van dijkringen niet altijd dé oplossing. H2O 2006, 21, 29-31 (in Dutch).
- UNEP, 2011, Towards a Green Economy: Pathways to Sustainable Development and Poverty Eradication, www.unep.org/greeneconomy
- Van der Biest, K.; Verwaest, T.; Vanneuville, W.; Reyns, J.; Mostaert, F. (2008). CLIMAR: evaluation of climate change impacts and adaptation responses for marine activities. Subdocument coastal flooding. Version 2.0. WL Rapporten, 814\_01. Flanders Hydraulics Research: Antwerp.
- Van der Biest, K.; Verwaest, T.; Reyns, J.; Mostaert, F. (2009a). CLIMAR: Deelrapport 2 -Kwantificatie van de secundaire gevolgen van de klimaatsverandering in de Belgische kustvlakte. Versie 2\_0. WL Rapporten, 814\_01. Waterbouwkundig Laboratorium: Antwerpen, België
- Van der Biest, K.; Verwaest, T.; Mostaert, F. (2009). CLIMAR: Section report 3 Adaptation measures to climate change impacts along the Belgian coastline. Version 2\_0. WL Rapporten, 814\_01. Flanders Hydraulics Research: Antwerp, Belgium
- Vanneuville W., Maddens R., Collard Ch., Bogaert P., De Maeyer Ph., Antrop M., (2006) Impact op mens en economie t.g.v. overstromingen bekeken in het licht van wijzigende hydraulische condities, omgevingsfactoren en klimatologische omstandigheden, studie uitgevoerd in opdracht van de Vlaamse Milieumaatschappij, MIRA, MIRA/2006/02, Gent. (Dutch)
- Verwaest T., Viaene P., Verstraeten, J. & Mostaert F., 2005. De zeespiegelstijging meten, begrijpen en afblokken. De Grote Rede 15, december 2005.
- Verwaest, T., Vanneuville, W., Peeters, P., Mertens, T., & Wolf, P. D. (2007). Uncertainty on coastal flood risk calculations and how to deal with it in coastal management. Case of the Belgian coastal zone. Flood Risk Assessment II: proceedings of a conference held by the Institute of Mathematics and its Applications (pp. 151-157).
- Verwaest, T.; Van der Biest, K.; Vanpoucke, Ph.; Reyns, J.; Vanderkimpen, P.; De Vos, L.; De Rouck, J.; Mertens, T. (2009). Coastal flooding risk calculations for the Belgian coast, in: McKee Smith, J. (Ed.) (2009).Proceedings of the 31st International Conference on Coastal Engineering 2008, Hamburg, Germany, 31 August to 5 September 2008. pp. 4193-4201
- Volckaert A. and R. De Sutter, 2009. CLIMAR Evaluation of climate change impacts and adaptation responses for marine activities. Subdocument: Sector Tourism. Report prepared in the framework of the CLIMAR project for the Belgian Science Policy, Contract SD/NS/01A, Arcadis Belgium, Gent, 75 pp.
- Vriesacker M., Platteau J. & Van Gijseghem D., 2007. Landbouw in zakformaat. Land- en tuinbouw in Vlaanderen 2007. Beleidsdomein Landbouw en Visserij, afdeling Monitoring en Studie Brussel.
- Willems, P. (2007). Extreme-waarden-analyse hoogwaterstanden te Oostende. Waterbouwkundig Laboratorium: Antwerpen. I, 14 pp.
- WLO (2006). Welvaart en Leefomgeving: een scenariostudie voor Nederland in 2040. Central Planning Bureau, Netherlands Environmental Assessment Agency and Spatial Planning Bureau (in Dutch).